Title: Non-European Union doctors in the National Health Service: why, when and how do they come to the United Kingdom of Great Britain and Northern Ireland?

Authors:

Jyothis T George (dritgeorge@yahoo.co.uk)
Kavitha S Rozario (kavitharoz@yahoo.com)
Jeffrin Anthony (jeffrinanthony@gmail.com)
Edward B Jude (edward.jude@tgh.nhs.uk)
Gerard A McKay (gerard.mckay@lanarkshire.scot.nhs.uk)

Version: 3 Date: 24 February 2007

Author's response to reviews:

Uploading on behalf of author.